\documentclass{article}
\usepackage{multirow,nopageno,fullpage}
\input{coursemacros}


\begin{document}
\head{Math 390 homework rubric}


\newcommand{\rrow}[3]{%
{\bf #1}: & \\
#2 \vspace{.75in} & 
\qquad\qquad / #3 \\ }

\begin{tabular}{|p{14.5cm}|c|}
\hline
\rrow{Mathematical correctness}
{Are the answers to the exercises correct?

(Easy question is worth 20 points, medium and hard are 15 points)
}{50}
\hline
\rrow{Mathematical formatting}
{Is the mathematical content presented in a sensible and well-formatted way? 

Are tables and pictures used properly and when appropriate?

Is it typed with \LaTeX?}{20}
\hline
\rrow{Writing style}
{Is the style of writing appropriate, with ideas presented formally and coherently? 

Are ideas presented in complete sentences and paragraphs, with spelling and grammatical errors minimal?

Did you include the statement of each exercise as written in the book? (don't bother copying over pictures)}{20}
\hline
\rrow{Difficulty}
{Do your three written exercises have approximately easy, medium, and hard difficulty?}{10}
\hline
{\bf Total}: & \qquad\qquad / 100 \\
\hline
\end{tabular}

\end{document}